\documentclass[11pt]{article}
\usepackage{pandora,euler}
\pagestyle{empty}

\begin{document}
\begin{center}
{\Large {Mathematics 2B, Winter 1998}}\\
\large Review Guide for Midterm I
\end{center}
\medskip

\begin{enumerate}

\item Area bounded between curves:  See homework from Section 5.8.
Also, can you prove that the area of a circle is $A=\pi r^2$?  Can you
do p.~530 no.~109?

\item Techniques of Integration:  Read the Chapter 9 Summary on page
528 and do as many \emph{diverse} problems from the Chapter~9 Review
as possible.  Skip the problems that ask about volumes, surfaces,
paths, and Newton's method---everything else is game.

\item Initial Value Problems:  Use the technique from the Friday 
(1/30) lecture.  Problems like the ones on p.~507 are possible.

\end{enumerate}

General notes:

\begin{itemize}

\item No multiple-choice questions, no calculators.  A table with
integral and trigonometric formulas will be supplied if necessary.

\item In most exam problems you will \emph{not} be told which method of
integration to use.

\item Get used to naming constants with letters, especially in the
context of initial value problems.

\item If you want to leave from the exam early, you must do so at least 10
minutes before the end of the exam.

\item Your neighbors in the exam room will have different questions to work
on.

\item Remember the \emph{Give Up} option.

\end{itemize}

\end{document}
